Title: Streptococcus gordonii septic arthritis: two cases and review of literature

Authors:

Jean Cyr Yombi (jean.yombi@uclouvain.be)
Leila Belkhir (leila.belkhir@uclouvain.be)
Sylvie Jocnkheere (sylvie.jonckheere@uclouvain.be)
Dunja Wilmes (dunja.wilmes@uclouvain.be)
Olivier Cornu (olivier.cornu@uclouvain.be)
Bernard Vandercam (bernard.vandercam@uclouvain.be)
Hector Rodriguez (hector.rodriguez@uclouvain.be)

Version: 5 Date: 23 June 2012

Author’s response to reviews: see over
To: Ms. Roselle Pangilinan  
Prof Shabir Madhi  
From: Dr J. C. Yombi  
Subject: MS: 2133637499647244

Dear Editor,

We thank you for giving us the opportunity to revise the manuscript MS: 2133637499647244 Streptococcus gordonii septic arthritis: two cases and review of literature.

Please find enclosed the revised version, modified according to your comments and specific style requirements. A consent section has been added. Authors’ e-mail addresses and specific contributions are specified. Grammar and spelling have been checked and corrected.

We thank you for these helpful comments and corrections and hope that the revised manuscript may be considered suitable for publication.

Yours sincerely,

Dr Jean Cyr Yombi